Author’s response to reviews

Title: Public Views of Acceptability of Perinatal Mental Health Screening and Treatment Preference: A Population Based Survey.

Authors:

Dawn A Kingston (dawn.kingston@ualberta.ca)
Sheila McDonald (shielaw.mcdonald@albertahealthservices.ca)
Suzanne Tough (suzanne.tough@albertahealthservices.ca)
Marie-Paule Austin (m.austin@unsw.edu.au)
Kathy Hegadoren (kmh3@ualberta.ca)
Gerri Lasiuk (glasiuk@ualberta.ca)

Version: 4  Date: 24 January 2014

Author’s response to reviews: see over
Jan. 23, 2014
Dr. Peter O’Donovan
Executive Editor
BMC Pregnancy and Childbirth
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Re: Re-submission of manuscript, Public Views of Acceptability of Perinatal Mental Health Screening and Treatment Preference: A Population Based Survey

Dear Dr. O’Donovan:

Please find enclosed our revised manuscript, Public Views of Acceptability of Perinatal Mental Health Screening and Treatment Preference: A Population Based Survey, by Kingston et al. Thank-you for the opportunity to resubmit the manuscript. In this version, we have added the tables following the reference list and removed them from the ‘supplementary materials’ files. In addition, we have added a statement regarding consent on page 7 following our description of ethics (‘Oral consent was obtained from each participant by the interviewer after the interviewer read a standard statement regarding the voluntary nature of his/her participation and confidentiality of the information collected in accordance with the Alberta Freedom of Information and Protection of Privacy Act.’)

With great appreciation,
Dr. Dawn Kingston, Ph.D, RN (Corresponding author)
Assistant Professor, Faculty of Nursing
Adjunct Assistant Professor, Faculty of Medicine, Dept of Obstetrics and Gynecology
5-258 Edmonton Clinic Health Academy,
University of Alberta
Edmonton, Alberta
11405-87th Avenue
T6G 1C9
dawn.kingston@ualberta.ca
Phone: 780-492-4731
Fax: 780-492-2551